ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Chew, Soo-Hong; Miao, Bin; Zhong, Songfa

Article Ellsberg meets Keynes at an urn

Quantitative Economics

Provided in Cooperation with: The Econometric Society

Suggested Citation: Chew, Soo-Hong; Miao, Bin; Zhong, Songfa (2023) : Ellsberg meets Keynes at an urn, Quantitative Economics, ISSN 1759-7331, The Econometric Society, New Haven, CT, Vol. 14, Iss. 3, pp. 1133-1162, https://doi.org/10.3982/QE2253

This Version is available at: https://hdl.handle.net/10419/296340

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by-nc/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Ellsberg meets Keynes at an urn

Soo Hong Chew

China Center for Behavioral Economics and Finance, Southwestern University of Finance and Economics and Department of Economics, National University of Singapore

Bin Miao

School of Economics, Renmin University of China

Songfa Zhong

Department of Economics, National University of Singapore and Department of Economics, New York University Abu Dhabi

Keynes (1921) and Ellsberg (1961) have articulated an aversion toward betting on an urn containing balls of two colors of unknown proportion to one with a 50–50 composition. Keynes views this as reflecting different preferences for bets arising from different sources of uncertainty. Ellsberg describes this as weighting the priors arising from the unknown urn pessimistically. In two experiments, we observe substantial links between attitude toward almost-objective uncertainty and attitudes toward multiple-prior uncertainties in terms of ambiguity and its corresponding compound risk. Our findings point to a shared component across domains of uncertainty and motivate the need for further theoretical development.

KEYWORDS. Ambiguity, source preference, maxmin expected utility, recursive utility, experiment.

JEL CLASSIFICATION. C91, D81.

1. INTRODUCTION

The proverbial urn has served a useful role as a canonical source of uncertainty since time immemorial. In probability and statistics, inferring the likelihood of drawing a ball from an urn with a known color composition is a basic problem, which has given rise to a rich range of studies spanning multiple disciplines. In an interesting twist evident in two seminal works 40 years apart, Keynes (1921) and Ellsberg (1961) have each come up

Soo Hong Chew: chew.soohong@gmail.com

Bin Miao: binmiao11@gmail.com

Songfa Zhong: zhongsongfa@gmail.com

© 2023 The Authors. Licensed under the Creative Commons Attribution-NonCommercial License 4.0. Available at http://qeconomics.org. https://doi.org/10.3982/QE2253

We would like to thank the audiences at DTEA, FUR, and RUD for their stimulating discussions and valuable comments. We thank the anonymous referees for comments that greatly improved the paper. We are grateful to David Freeman, Yoram Halevy, Collin Raymond, and Peter Wakker for their helpful comments and suggestions. Chew and Zhong thank the National University of Singapore and Singapore Ministry of Education for financial supports. Miao acknowledges the support of Beijing Social Science Fund Project (21DTR007), NSF China (Grants 71922014, 72131003). Chew acknowledges the support of NSF China (Grant 72033006).

with the idea of an urn containing colored balls of unknown proportions. Intriguingly, they offer distinct perspectives on the same unknown urn in a decision-making setting.

In *A Treatise on Probability* (1921, p. 83), Keynes proposes a thought experiment between betting on an urn with equal proportions of white and black balls over betting on another urn in which the proportion of each color is unknown—and points out that the probability of drawing a white (black) ball from either urn is evidently 1/2, but the "weight of argument" is greater for the known urn leading possibly to a preference for the known bet. He asks, "if two probabilities are equal in degree, ought we, in choosing our course of action, to prefer that one which is based on a greater body of knowledge?" In contrast, Ellsberg (1961) observes that the unknown bet encompasses a range of possible priors and hypothesizes that the known bet is favored for those who are pessimistic in weighing the less favorable priors disproportionately. Ellsberg also coined the term ambiguity aversion to refer to a preference for the known bet over the unknown bet.

As it turns out, while the literature following Ellsberg (1961) has been vibrant, there has not been as much follow-up work on Keynes (1921). The Ellsbergian multiple-prior perspective gives rise to two directions in modeling ambiguity attitude. One direction is to assign all weights to the worst prior as axiomatized in Gilboa and Schmeidler's (1989) model, which delivers the lowest among the possible expected utilities.¹ Another direction can be traced to Becker and Brownson (1964) who offer a parallel view of the unknown bet as a subjective compound lottery over the range of possible priors (which can be further reduced to an even-chance simple lottery). As such, ambiguity aversion corresponds to a preference for the even-chance simple lottery from the known urn over the induced subjective compound lottery from the unknown urn in violation of the reduction of compound lottery axiom (RCLA).² Adopting this two-stage view, Segal (1987) models ambiguity aversion by applying a probability weighting function that overweights the less favored priors (in terms of delivering lower stage-2 utilities). This theoretical relation between the two different approaches to modeling ambiguity aversion, both involving multiple priors, motivates Halevy's (2007) experimental study on the link between attitudes toward ambiguity and objective compound lottery in which the distribution of priors is objectively given as uniform. Halevy's (2007) finding of a tight link between these two attitudes lends support to the compound lottery view and the corresponding recursive modeling approach of Segal (1987) and subsequently Klibanoff, Marinacci, and Mukerji (2005), Ergin and Gul (2009), Seo (2009).

Echoing Keynes, Smith (1969) arrives independently at the view that both the known and unknown bets are in effect 50–50 when subjects do not prefer one color over the other for balls drawn from either urn. He proposes to make use of two distinct von Neumann–Morgenstern utility functions so that a known bet can be worth more than

¹Ghirardato, Maccheroni, and Marinacci (2004) offer a more moderate model of pessimism through a convex combination of the best and the worst prior. Gajdos, Hayashi, Tallon, and Vergnaud (2008) axiomatize the pessimism decision rule in Ellsberg (1961), which takes a convex combination of the worst prior and an "estimated' half/half prior. Schmeidler (1989) and Gilboa (1987) can capture the intuition of pessimism directly by assigning a greater decision weight than 1/2 to the losing event.

²RCLA requires that a compound lottery is indifferent to its reduction to a simple lottery with the same overall probabilities and outcomes.

an unknown bet. This can be accomplished as long as the utility function associated with risks arising from the unknown urn is more concave than that associated with risks from the known urn.³ Building on Keynes' perspective, Fox and Tversky (1995) consider uncertainty arising from natural events such as the price of a stock and temperature of a city. In their experiment, subjects in the Bay Area value separately two pairs of bets with potentially similar likelihoods of winning: one pair is based on whether the temperature one week from today in San Francisco is above or below 60°F; another pair is based on whether the temperature one week from today in Istanbul is above or below 60°F. Surprisingly, they observe that the lower-valued bet on the San Francisco temperature is worth more than the higher-valued bet on the Istanbul temperature. This has led Fox and Tversky to originate the concept of source preference, referring to a preference for betting on sources of uncertainty, such as a known urn or a familiar city both involving greater knowledge. This source preference perspective prompts Chew and Sagi (2008) to offer an axiomatization more generally so that the decision maker can have distinct attitudes toward risks arising from different sources of uncertainty that induce the same distribution, whether they are from an urn, from a deck of cards, or from natural events such as the temperature of a city.

Following Halevy (2007), there has been several experimental studies of the link between ambiguity attitude and compound risk attitude (e.g., Abdellaoui, Klibanoff, and Placido (2015), Chew, Miao, and Zhong (2017), Gillen, Snowberg, and Yariv (2019), Dean and Ortoleva (2019)). This strand of findings reinforces the Ellsbergian multiple-prior perspective and at the same time points to the need to study whether these two attitudes relate to source preference arising from the Keynesian perspective. The present paper addresses this gap in the literature via Experiment 1, which examines attitudes toward three types of uncertainty besides attitude toward the benchmark 50–50 known risk from betting on the color of a card drawn from a deck of 20 cards with 10 red and 10 black.

Ambiguity: Betting on the color of a card drawn from a deck of 20 cards each being either red or black. The composition is (partially) unknown.

Compound risk: Betting on the color of a card drawn from a deck of 20 cards each being either red or black. The composition is objectively determined by a random device.

Natural-event uncertainty: Betting on the trailing decimal digit (odd versus even) of the highest temperature in a city on a pre-specified historical date.

Our experimental design encompasses three ambiguity lotteries, two compound lotteries, and two natural-event lotteries based on two different cities. All these lotteries deliver a half chance of winning in the spirit of Keynes. Notice that our choice of naturalevent uncertainty is distinct from that in Fox and Tversky (1995) in that the events of odd/even in the temperature bets are 50–50 in an "almost objective" sense (see Section

³Smith (1969) attempts to model preference for the known urn bets by arguing that the "known" utility is cardinally higher than the "unknown" utility, but this is not legitimate given that vNM utility functions are unique up to affine transformations. In raising the possibility of adopting two distinct vNM utility functions for the two sources of uncertainty, Pratt's (1964) relative concavity condition can be applied in order to arrive at the result of the known bet being worth more than the unknown bet.

II for a detailed discussion). We adopt a within-subject design with more than 2000 undergraduate subjects in Singapore. Using a choice-list mechanism, we elicit the certainty equivalent (CE) for each lottery.

We infer subjects' attitude toward ambiguity (compound risk/natural event) by its premium, which is given by the difference between the CE of the benchmark 50–50 risky lottery and that of ambiguity (compound risk/natural event) lottery. In other words, there is neutrality (aversion) toward a certain type of uncertainty when the corresponding premium is zero (positive).⁴ The majority of our subjects exhibit distinct attitudes toward the three types of uncertainty. Three-fourths of the subjects exhibit nonneutrality toward ambiguity as well as compound risk and two-thirds exhibit nonneutrality toward the natural-event uncertainty of a city temperature. Besides replicating the significant association between ambiguity attitude and compound risk attitude in the literature, we find that natural-event uncertainty attitude is closely linked to both attitudes toward ambiguity and compound risk. Subjects who are neutral toward the odd-even bet on temperature are more likely to be ambiguity neutral and more likely to reduce compound lottery. We observe significant correlations for premium across the three types of uncertainty and, as anticipated, the correlation coefficient between ambiguity and compound risk is substantially higher than the other two correlation coefficients (Table 1). In individual type analysis, we find that 65.3% of the subjects are nonneutral toward all three types of uncertainty, and 5.7% of the subjects are neutral toward all three. For the rest of subjects, 20.4% are nonneutral toward ambiguity and compound risk, and are neutral toward natural-event uncertainty. These three groups of subjects account for 91.3% of the subjects, with the rest of 8.7% exhibit other behavioral patterns. Moreover, the observed patterns do not differ substantially between subjects with different cognitive ability measured by Raven's progressive matrices. Finally, we show that the observed results are also robust with respect to replication using more than 1000 undergraduate subjects in Beijing.

In the literature of decision-making under risk and uncertainty, it is commonly observed that risk and ambiguity attitudes depend on both outcome and likelihood. A natural question arises as to whether the links observed in Experiment 1 may also exhibit

	Ambiguity	Compound risk
Compound risk	0.825	
Natural event	0.580	0.657
	(0.008)	(0.011)

TABLE 1. Correlations across three types of uncertainty.

Note: The correlation is estimated using the obviously related instrumental variables (Gillen, Snowberg, and Yariv (2019)) with standard errors.

⁴We focus on the comparisons among the premia for three types of uncertainty in this current paper. When it comes to the comparison between uncertainty premium and risk premium, caution needs to be taken since one may need to first separate uncertainty premium into belief-related premium and curvaturerelated premium, which could be model specific. See Baillon, Bleichrodt, Li, and Wakker (2021) for a recent discussion. dependence on outcome and likelihood. To address this question, we conduct Experiment 2 in which we examine four lotteries in line with Tversky and Kahneman's (1992) fourfold pattern of risk attitude-moderate prospect (50% chance of winning), longshot prospect (10% chance of winning), moderate hazard (50% chance of losing), longshot hazard (10% chance of losing)—in addition to a mixed lottery (50% chance of winning or losing). For each of the five lotteries, we consider again the three types of uncertainty in Experiment 1 besides known risk. To implement our four-plus-one design for known risk, we make use of a deck of 10 cards numbered 1 to 10. Even-chance (10% chance) corresponds to betting on parity (the number) of a card drawn. For ambiguity, the corresponding lotteries are generated by betting on a deck of 10 cards with unknown composition of numbers from 1 to 10. For compound risk, the deck of 10 cards to be bet on is constructed with each card independently drawn from a known deck of 10 cards numbered from 1 to 10. For natural-event uncertainty, subjects are to bet on the second decimal digit of a stock index at market closing. Overall, we find strong support for the robustness of the close links across the attitudes toward all three types of uncertainty in relation to risk-the correlations among attitudes toward ambiguity, compound risk, and natural-event uncertainty are on average 0.58.

Our overall results shed light on decision theoretic models under uncertainty. In particular, our inclusion of natural-event uncertainty helps link the two perspectives on ambiguity—multiple priors and source preference—which differentiate source-based models from those reflecting the multiple-prior perspective. At the same time, where natural-event uncertainty is concerned, the observed nonneutrality toward almostobjective events provides support for source preference in distinguishing among identically distributed risks. Together with the observed close links between attitude toward natural-event uncertainty and ambiguity attitude as well as compound risk attitude, our overall findings point to the need to extend the current recursive models by incorporating within-stage source preference.

Our results also contribute to the understanding on the experimental and empirical studies on risk attitudes across different domains. Weber, Blais, and Betz (2002) show that risk attitudes are not consistent across five domains including financial decisionmaking, health/safety, recreational, ethical and social decisions, and propose that risk attitudes are domain specific. Interestingly, the domains in Weber, Blais, and Betz (2002) differ in terms of whether risks are gain-oriented or loss-oriented, whether the odds are moderate or skewed as well as whether risks are arising from different sources of uncertainty. Barseghyan, Prince, and Teitelbaum (2011) examine deductible choices across domains of auto and home insurance and reject the null of fully domain-general risk aversion. Einav, Finkelstein, Pascu, and Cullen (2012) observe a strong correlation for five employer-provided insurance coverage decisions, and a considerably weaker relationship between insurance decisions and 401 (k) asset allocation. In an experimental study across 30 countries, Vieider et al. (2015) show that risk and uncertainty attitude correlate not only within decision contexts or measurement methods but also across contexts and methods. These findings are consonant with the results of our experimental study of attitudes toward risk across gains and losses and with different likelihoods arising from different sources of uncertainty, pointing to a shared component in decision-making across a range of domains of uncertainty.

The rest of the paper is organized as follows. Section 2 presents the theoretical background. Section 3 details our experimental design and Section 4 presents the main results. We discuss the implications of our findings in Section 5. Finally, Section 6 concludes. The Online Supplementary Material (Chew, Miao, and Zhong (2023)) include the supplementary tables and figures, experimental instructions, and the replication files (data and code).

2. Theoretical background

Denote by *S* the set of all possible states in the experiment. Consider four types of uncertainty: risk (*R*), ambiguity (*A*), compound risk (*C*), and almost-objective naturalevent uncertainty (*N*), and each type $i \in \{R, A, C, N\}$ partitions the state space into two equally likely events: $\{E_i, E_i^c\}$. Let h_i be the bet on event E_i , which pays *w* if event E_i occurs and 0 otherwise. The two events are equally likely in the sense that a decision maker is indifferent between betting on E_i and betting on E_i^c . Denote by **R**, **A**, **C**, and **N**, the lotteries induced by the corresponding act h_i , and $CE_{\mathbf{R}/\mathbf{A}/\mathbf{C}/\mathbf{N}}$ the certainty equivalent of each lottery.

Two points are noteworthy. First, the compound lottery **C** in fact corresponds to the "higher-order" act induced by h_C , which maps from $\Delta(S)$, the set of probability distributions on *S* (priors) to $\Delta(Z)$, the set of probability distributions on outcomes (simple lotteries). Second, the natural-event uncertainty considered here is "*almost-objective*" (Machina (2004)). For example, consider the uncertainty associated with the trailing digit of a city's temperature being either odd or even; it is almost objective in the sense that for any continuous random variable on the thermometer space with a smooth density function, the probability of odd approaches 0.5 asymptotically as the location of the decimal (e.g., 8th, 9th, etc.) goes to infinity. The intuition is clear. First, partition the state space into equal-length intervals, for example, $\{..., (0, \frac{1}{m}], (\frac{1}{m}, \frac{2}{m}], ...\}$. It can be shown that the (subjective) probability of the union of all the left-half of each interval, $\bigcup_n(\frac{n}{m}, \frac{n+0.5}{m}]$, converges to 0.5 as *m* grows large regardless of the prior distribution as long as its density is smooth.

Using **R** as the benchmark, we infer decision-maker's attitudes toward **A**, **C**, and **N**, as well as the three pairwise correlations of these attitudes. Consider the benchmark subjective expected utility (SEU) model, the utilities for different lotteries are as follows:

$$U_{\rm SEU}(\mathbf{R}/\mathbf{A}/\mathbf{C}/\mathbf{N}) = \mathbf{E}_p u \circ h_i,$$

where $p \in \Delta(S)$ denotes the (subjective) prior. SEU, or more generally probabilistic sophistication (Machina and Schmeidler (1992)), entails $p(E_i) = p(E_i^c) = 0.5$ in each lottery. Thus, SEU predicts that four lotteries $\mathbf{R}/\mathbf{A}/\mathbf{C}/\mathbf{N}$ have the same CEs, and hence generate no correlation across attitudes toward ambiguity, compound risk, and naturalevent uncertainty.

In the sequel, we discuss the predictions of a number of models under uncertainty in relation to the three attitudes as well as the interplay among these attitudes. All these models have been developed to accommodate ambiguity nonneutrality, and we differentiate among them according to whether they adopt an *multiple-prior perspective* by incorporating the possibility of multiple priors, or the *source preference perspective*, which insists on a single prior yet allows distinct attitudes toward different types of uncertainty.

2.1 Ellsbergian perspective

One-stage modeling incorporating multiple priors In his 1961 paper, Ellsberg advocates a "conservatism" rule in assigning a positive weight to the worst prior in the ambiguous bet. Gilboa and Schmeidler's (1989) *maxmin expected utility* model (MEU), widely acknowledged to be the representative model under the Ellsbergian perspective, represents the extreme case of assigning a weight of 1. Under MEU, the utility for ambiguity lottery **A** is given below:

$$U_{\mathrm{MEU}}(\mathbf{A}) = \min_{p \in \Pi} \mathbf{E}_p u \circ h_A,$$

where $\Pi \subset \Delta(S)$ is a convex set of priors with $\min_{p \in \Pi} p(E_A) = \min_{p \in \Pi} p(E_A^c)$. MEU exhibits ambiguity aversion $CE_{\mathbf{R}} \ge CE_{\mathbf{A}}$ when $p(E_A)$ admits nonunique values in Π .

In relation to compound risk attitude, the MEU model axiomatized in Gilboa and Schmeidler (1989) predicts compound risk neutrality $CE_{\mathbf{R}} = CE_{\mathbf{C}}$, since it adopts the Anscombe–Aumann framework, which incorporates RCLA.⁵ In contrast, the axiomatization of MEU in a Savagian domain (Casadesus-Masanell, Klibanoff, and Ozdenoren (2000), Alon and Schmeidler (2014)) makes no predictions concerning how compound lotteries may be evaluated. With regard to natural-event lottery **N**, MEU predicts neutrality toward almost-objective uncertainty given that $p(E_N)$ approaches 0.5 asymptotically.

In sum, MEU can generate ambiguity nonneutrality, but fails to predict nonneutral attitudes toward either compound risk or almost-objective natural event and, therefore, does not generate correlations among different uncertainty attitudes.⁶

Two-stage modeling incorporating multiple priors When incorporating multiple priors, an alternative modeling approach (Segal (1987)) views ambiguity as a distribution over the set of possible priors. Formally, such a two-stage modeling approach associates the ambiguity lottery **A** with a stage-1 prior μ_A , a distribution on $\Delta(S)$. It follows that for a given act h_A , μ_A induces a subjective compound lottery, which can be evaluated recursively. There are two ways to form a recursive evaluation in the literature.

⁵RCLA is implicit in the assumption of reversal of order in Anscombe and Aumann (1963). See Seo (2009) for a detailed discussion.

⁶The CEU model axiomatized in Gilboa (1987) and Schmeidler (1989) generalizes SEU with a capacity function ν , a nonadditive extension of a probability measure that maps events into the unit interval and is monotonic in terms of inclusion. In relaxing additivity, CEU is compatible with both ambiguity aversion and ambiguity tolerance without restrictions on the capacities $\nu(E_A)$ and $\nu(E_A^c)$. It, in fact, accommodates MEU given a convex ν .

For compound risk **C**, the observations of MEU also apply to CEU models including those adopting the Anscombe–Aumann framework (Schmeidler (1989)) and those adopting a Savagian domain (Gilboa (1987), Wakker (1987)). In addition, the almost objectivity of events odd/even in **N** again entails $CE_R \simeq CE_N$ in the CEU model.

Segal (1987, 1990) applies a *recursive rank-dependent utility* specification (RRDU) to evaluate **A** as follows:

$$U_{\text{RRDU}}(\mathbf{A}) = \int_{\Delta(S)} \int_{S} (u \circ h_A) \, df(\mathbf{P}) \, df(\mathbf{M}_A),$$

where \mathbf{M}_A (**P**) is the cumulative distribution function derived from μ_A (*p*), and *f* (*u*) a common probability weighting function (utility function) applied to both stages.

An alternative model is to apply expected utility preference recursively in evaluating the compound lottery, which is accomplished in the *recursive expected utility* model (REU) of Klibanoff, Marinacci, and Mukerji (2005) and Seo (2009) as follows:

$$U_{\rm REU}(\mathbf{A}) = \mathbf{E}_{\mu_{\mathbf{A}}} v(\mathbf{E}_p u \circ h_A),$$

where v and u are stage-1 and stage-2 utility functions, respectively.

Both RRDU and REU can exhibit ambiguity nonneutrality in violating RCLA for the induced subjective compound lottery. In particular, Segal (1987) shows that a convex f in RRDU can be compatible with ambiguity aversion $CE_R \ge CE_A$, while a concave v in REU is shown to be compatible with ambiguity aversion in Klibanoff, Marinacci, and Mukerji (2005).⁷ For compound lottery **C**, μ_C is objective and both models can predict compound-risk aversion: $CE_R \ge CE_C$, under the same conditions as that for ambiguity aversion. Hence, both models are compatible with positive correlation between compound risk attitude and ambiguity attitude.⁸ When considering almost objective natural-event lottery **N**, μ_N collapses to an atom, and it follows that both models predict natural-event uncertainty neutrality, that is, $CE_R \simeq CE_N$, and cannot generate any correlation between ambiguity/compound risk attitude and natural-event uncertainty attitude.

2.2 Keynesian perspective

One-stage source Fox and Tversky (1995) provide evidence in support of Keynes' source perspective, prompting Chew and Sagi (2008) to axiomatize source preference directly in terms of probabilistic sophistication on endogenous smaller families of events. In their model, ambiguity lottery **A** is viewed as an alternative source of uncertainty in addition to **R** and **N**. Note that we shall be using the term "*source of uncertainty*" only when referring to source preference models in the sequel. The Chew–Sagi model displays flexibility

⁷Halevy and Feltkamp (2005) link ambiguity aversion with aversion to mean-preserving spreads in an environment of bundled risks and deliver similar predictions as those of REU.

⁸Note that we focus on the functional forms of recursive models when discussing their predictions on ambiguity attitude, compound risk attitude, as well as the correlation between the two. Formally, the REU model axiomatized in Klibanoff, Marinacci, and Mukerji (2005) adopts a Savagian domain and makes no predictions on its behavior in the domain of compound risks. In contrast, Seo's (2009) approach of distinguishing distributions over acts from the statewise mixture of acts gives rise to ambiguity and compound risk in a unified domain and is able to link ambiguity nonneutrality with non-RCLA behavior for compound risk. See Halevy (2007, pp. 515–517) for a related discussion on how various recursive models can(not) differentiate between objective and subjective stage-1 priors.

in evaluating **R**, **A**, and **N** with local source-specific utility functionals:

$$U_{\rm CS}(\mathbf{R}/\mathbf{A}/\mathbf{N}) = \mathbf{U}^{\iota}(p, h_i),$$

where \mathbf{U}^i can admit different (expected or nonexpected) specifications for different sources of uncertainty even when the prior *p* assigns the same probabilities to events $E_R/E_A/E_N$. In adopting a source-dependent probability weighting function using rankdependent utility (Quiggin (1982)), Abdellaoui, Baillon, Placido, and Wakker (2011) offer a more specific form of the Chew–Sagi model:

$$\mathbf{U}^{i}(p,h_{i}) = \int_{S} (u \circ h_{i}) \, df^{i}(\mathbf{P}),$$

where f^i is a source-dependent probability weighting function, and **P** the cumulative distribution function derived from *p*. Notably, Chew–Sagi admits RCLA for compound risk and predicts $CE_R = CE_C$. Given the flexibility in choosing **U**^{*i*} for different sources of uncertainty, Chew–Sagi can produce correlation between ambiguity attitude and natural-event uncertainty attitude, but fails to link compound risk attitude with the attitude toward ambiguity or with attitude toward natural-event uncertainty.

Two-stage source Nau (2006) distinguishes the uncertainty arising from two exogenously given "issues" (sources) and applies distinct expected utility preferences across the two different issues. By interpreting one issue as the first-order prior and the other as the second-order prior, Nau's representation coincides with REU. Ergin and Gul (2009) axiomatize a more general second-order probabilistic sophistication representation (SPS), which incorporates possibly distinct NEU preferences across different "issues."⁹

Specifically, the utility for ambiguity lottery A under SPS is as follows:

$$U_{\text{SPS}}(\mathbf{A}) = \mathbf{V}(\mu_A, \mathbf{U}(p, h_A)),$$

where **U** is the stage-2 utility functional evaluating act h_A under a given stage-2 prior p, and **V** the stage-1 utility functional that aggregates different stage-2 utilities according to the stage-1 prior μ_A . In SPS, **V** and **U** both admit probabilistic sophistication, and hence SPS incorporates REU and RRDU as special cases. It follows that both Nau (2006) and SPS can exhibit ambiguity nonneutrality and non-RCLA, as well as positive correlation between ambiguity attitude and compound risk attitude.

When it comes to **N**, Nau (2006) and SPS in general can treat risk and almostobjective uncertainty as two "issues" and deliver distinct evaluations between **R** and **N**. This notwithstanding, both models are more restrictive than Chew–Sagi in that they can deal with up to two exogenous sources of uncertainty. Therefore, upon interpreting the two sources as priors at different stages, Nau (2006) and SPS fail to generate naturalevent nonneutrality and deliver the same predictions as RRDU and REU.

⁹Both models work on a product state space $S_1 \times S_2$, where S_1 and S_2 are assumed to be two exogenous issues of uncertainty and are free to admit the interpretation of priors at different stages. Broadly speaking, the REU model as in Klibanoff, Marinacci, and Mukerji (2005) applies distinct expected utility preferences for priors at different stages, and hence can also be regarded as a two-stage source model.

		Α	С	Ν	A–C	A–N	C–N
Ellsbergian	Pessimism	Nonneutral	Neutral	Neutral	N	N	N
	Recursive	Nonneutral	Nonneutral	Neutral	Y	N	N
Keynesian	One-stage source	Nonneutral	Neutral	Nonneutral	N	Y	N
	Two-stage source	Nonneutral	Nonneutral	Neutral	Y	N	N

TABLE 2. Theoretical predictions.

Note: This table summarizes the qualitative predictions of different modeling approaches regarding attitude toward each type of uncertainty, as well as the pairwise correlations among those attitudes. "Neutral" ("Nonneutral") for a specific modeling approach means it is incompatible (compatible) with nonneutral attitude toward certain type of uncertainty. When marked in grey, "Neutral"/'Nonneutral" includes the case that some specific model within the approach makes no predictions concerning the corresponding uncertainty attitude, depending on the axiomatization framework. Similarly, a "Y" for a specific approach means it can generate correlations between two corresponding uncertainty attitudes, while a "N" means it either makes no prediction concerning or is incompatible with correlations between different uncertainty attitudes.

2.3 Summary

Table 2 summarizes the theoretical predictions of different models in terms of attitudes toward the three types of uncertainty (ambiguity/compound risk/nature-event uncertainty), as well as pairwise correlations among these attitudes.

Except for SEU, all models considered can exhibit nonneutrality toward **A**. Taking compound risk **C** into consideration, the observed "*known*" link (Halevy (2007), Abdellaoui, Klibanoff, and Placido (2015), Chew, Miao, and Zhong (2017)) between ambiguity attitude and compound risk attitude (**A**–**C**) helps distinguish between the two approaches both adopting the multiple-prior perspective—those with built-in pessimism by overweighting the worst prior to accommodate ambiguity aversion, and those assuming higher-order distributions on the set of priors coupled with non-RCLA to deliver ambiguity nonneutrality. Specifically, MEU, as a representative of the former, permits RCLA (or makes no predictions concerning how compound lottery should be evaluated, depending on the axiomatization framework), and thus is unable to generate such a known link. In contrast, recursive models, including RRDU and REU, view ambiguity lottery as subjective compound lottery, and hence are compatible with the known link.

Considering additionally almost-objective natural-event uncertainty helps distinguish further models adopting the *multiple-prior* or *source preference* perspective. In particular, all multiple-prior models fail to generate (nonnegligible) difference between an objective risk **R** and an "almost" objective lottery **N**. It follows that they cannot generate links between natural-event uncertainty attitude and either ambiguity attitude (**A**–**N**) or compound risk attitude (**C**–**N**). In contrast, the Chew–Sagi source model, by treating different equally-likely events distinctly, is able to generate a link between ambiguity attitude and natural-event uncertainty attitude (**A**–**N**), but fails to deliver the other two possible links (**A**–**C** and **C**–**N**), given that it admits RCLA. Notably, the Nau or SPS model differentiates stage-1 and stage-2 priors as distinct sources of uncertainty. Consequently, their predictions coincide with those of the recursive multiple-prior models.

Beyond moderate prospect The preceding summary focuses on the qualitative predictions of different models, which can be extended naturally to loss outcomes as well as skewed winning and losing events. For example, consider a general setting in which each type of uncertainty $i \in \{R, A, C, N\}$ partitions the state space into *m* equally-likely events $\{E_i^1, E_i^2, \ldots, E_i^m\}$. Here, a bet on a particular event $\bigcup_{k \in K \subset \{1, 2, \ldots, m\}} E_i^k$ can induce different kinds of lotteries involving moderate (or skewed) chance of winning *w* (or losing *w'*). No matter which kind of lottery we consider, it remains valid that models adopting the Ellsbergian perspective fail to deliver nonneutrality toward almost-objective natural-event uncertainty, and thus are unable to generate links between natural-event uncertainty attitude and ambiguity attitude or compound risk attitude. In contrast, models adopting the Keynesian perspective, including the Chew–Sagi model, treat ambiguity and natural-event uncertainty as distinct sources of uncertainty in parallel with risk. While they are compatible with nonneutrality toward natural-event uncertainty, they fail to accommodate compound risk nonneutrality as well as the links between compound risk attitude and ambiguity attitude or natural-event uncertainty, they fail to accommodate and provide the set of uncertainty attitude. In contrast, models adopting the models adopting the keynesian perspective, including the Chew–Sagi model, treat ambiguity and natural-event uncertainty as distinct sources of uncertainty in parallel with risk. While they are compatible with nonneutrality toward natural-event uncertainty, they fail to accommodate compound risk nonneutrality as well as the links between compound risk attitude and ambiguity attitude or natural-event uncertainty attitude.

In sum, none of the existing models can generate nonneutrality toward all three types of uncertainty, nor pairwise links between attitudes toward the three types of uncertainty, regardless of the underlying lottery considered.¹¹ This observation provides the foundation of our experimental design, as illustrated in the subsequent section.

3. Experimental design

3.1 Experiment 1

Experiment 1 includes one risk lottery, three ambiguity lotteries, two compound lotteries, and two natural-event lotteries as summarized in Table 3. For risk lottery **R**, subjects bet on the color of a card randomly drawn from 20 cards with 10 red and 10 black, and they are explicitly told that the winning probability is 50%. The full ambiguity lottery **A** is the same as that in the Ellsberg paradox, and we include two more partial ambiguity lotteries, that is, interval ambiguity **A1** and disjoint ambiguity **A2** in which the partial information about the composition of the deck is given to the subjects. The full compound lottery **C** is implemented as follows. One ticket is drawn randomly from a bag containing 21 tickets numbered from 0 to 20, and the number drawn subsequently determines the number of red cards in the deck with the remaining cards black. Similarly, for the *p-q* compound lottery **C1**, one number is drawn from an envelope containing 8 tickets

¹¹In Section 5, we discuss extensions of existing source models, in terms of delivering nonneutral attitude toward all three sources of uncertainty as well as pairwise correlations among them.

¹⁰While it is straightforward that the current predictions for even-chance lottery can be extended to incorporate other kinds of lotteries, it remains a question whether or not different models are compatible with ambiguity/compound risk/natural-event aversion or seeking behavior. For example, consider skewed compound risk attitude under two-stage models, Segal (1987) and Dillenberger and Segal (2017) show that RRDU can generate either skewed compound-risk aversion or seeking, depending on the functional forms at each stage as well as the distribution of stage-1 priors.

The question is also related to the well-known fourfold pattern for risk—risk aversion for moderate prospect and longshot hazard as well as risk seeking for longshot prospect and moderate hazard. While fourfold pattern of risk has been extensively explored since the seminal works of Kahneman and Tversky (1979) and Tversky and Kahneman (1992), the experimental evidence of fourfold pattern for attitude toward ambiguity (or more generally attitudes toward different types of uncertainty), has been limited and mixed. As the current paper focuses on the links among attitudes toward different types of uncertainty, we would like to leave the question for future study.

	Lottery	Implementation
R	Risk	A deck of 20 cards with 10 red and 10 black.
A	Full ambiguity	A deck of 20 cards with no information about the composition of red and black.
A1	Interval ambiguity	A deck of 20 cards with the number of red (black) cards between 5 and 15.
A2	Disjoint ambiguity	A deck of 20 cards with the number of red (black) cards either between 0 and 5 or between 15 and 20.
С	Full compound risk	A deck of 20 cards with 21 possible compositions of red and black being equally likely.
C1	<i>p-q</i> compound risk	A deck of 20 cards with 5/8 (3/8) chance of containing 16 (0) red cards and 4 (20) black cards.
N N1	Natural event (home) Natural event (foreign)	Trailing digit of home city highest temperature on a given historical date. Trailing digit of foreign city highest temperature on a given historical date.

TABLE 3. Summary of lotteries used in Experiment 1.

numbered 1 to 8. If the number drawn from is 1 to 5 (6 to 8), the deck will have 16 (0) red cards and 4 (20) black cards. In both compound lotteries, subjects bet on the color of a card drawn from the deck before drawing the ticket, and the winning probability of betting on red (black) can be reduced to 50%. In natural-event lottery **N**, subjects bet on the trailing decimal digit (odd/even) of the temperature in their home city (Singapore) on a randomly selected historical date. Correspondingly, lottery **N1** is based on the temperature of a foreign city (Istanbul). We have included multiple lotteries for the three types of uncertainty besides the baseline known risk. This enables us to apply the method of Obviously Related Instrumental Variables (ORIV, Gillen, Snowberg, and Yariv (2019)) to conduct more efficient correlation analyses, which we elaborate further in Section IV.

In our experiment, subjects are allowed to choose the color of the card or parity of city temperature to bet, and the psychology of suspicion is less of a concern.¹² Subjects receive a winning prize SGD60 (about USD43) for a correct bet (guess) in each lottery, and nothing otherwise. To elicit the certainty equivalent (henceforth CE) of each lottery, we use a choice-list design with subjects choosing between a given lottery and a sure amount ranging from SGD15 to SGD35 (see Appendix B for experimental instructions). In general, subjects will have a single switching point, that is, choosing the lottery when its CE is higher than the sure amount and switch to the sure amount as it increases to the point that is higher than the CE of the lottery. Thus, the switch point serves as a proxy for CE of the lottery.

The eight choice-lists are implemented along with the other ten tasks examining risk preference and time preference. Subjects complete the decision-making tasks without feedback. To incentivize participation, in addition to a SGD35 show-up fee, we adopt the random incentive mechanism (RIM), paying each subject based on one of her randomly selected decisions in the experiment. The use of RIM to elicit valuations of lotteries has

¹²Subjects may suspect that the experimenter manipulates the composition of the cards against them when they are not allowed to choose the event to bet. Such suspicion may increase the observed ambiguity aversion (Berger and Tymula (2022)).

triggered debates from of both theoretical and experimental perspectives.¹³ We adopt RIM with the price list as it enables us to examine the potential links across different types of uncertainty.

In addition to the decision-making tasks, we include a test of cognitive ability using Raven's standard progressive matrices, one of the most popular measures of analytic intelligence. The test consists of five increasingly difficult parts of 12 questions each. In each question, subjects are asked to identify a missing element to complete a visual pattern. While we do not impose a time limit, we state that people tend to finish the test in 20–45 minutes. In the literature, higher performance on Raven's test has been found to be associated with fewer Bayesian updating errors (Charness, Rustichini, and Van de Ven (2011)), more accurate beliefs (Burks, Carpenter, Goette, and Rustichini (2009)), and increased likelihood of choosing strategies that are closer to Nash equilibrium and converge faster to equilibrium play (Gill and Prowse (2016)). The inclusion of Raven's test enables us to examine a possible relationship between cognitive ability and attitudes toward different types of uncertainty.

We have recruited a cohort of 2066 undergraduate students from the National University of Singapore who participated in a study on the biological basis of decisionmaking (53.0% female; mean age: 21.4). Following the Singapore cohort, an additional cohort of 1181 Han Chinese are recruited from several universities in Beijing (48.4% female; mean age: 21.5). The instructions and procedures are the same for the Singapore and Beijing experiments, except for the following points. First, the prize of the lotteries is changed to RMB240 (about SGD50). Second, for the natural-event lotteries, the home city is Beijing and the foreign city is Tokyo. Last, the written and oral instructions and decision-making tasks are in Chinese instead of English. All subjects provide written informed consent and the experiment was approved by the Institutional Review Board at the National University of Singapore.

3.2 Experiment 2

Experiment 2 goes beyond even-chance prospects and considers lotteries that vary in both outcome (gains versus losses) and likelihood (moderate versus longshot). Specifically, we consider five baseline risk lotteries: (1) moderate prospect: 50% chance of winning RMB100 (about USD14.54), otherwise 0; (2) moderate hazard: 50% chance of losing RMB50, otherwise 0; (3) longshot prospect: 10% chance of winning RMB100, otherwise

¹³RIM generates an objective compound lottery and may contaminate inferences of risk and ambiguity attitudes under certain circumstances. A sufficient condition for the validity of RIM is isolation (Kahneman and Tversky (1979)). See, for example, Wakker (2007) for a general discussion. In the domain of ambiguity, Bade (2015), Azrieli Y., Chambers, and Healy (2018), Baillon, Halevy, and Li (2022a), among others argue that RIM may not be incentive compatible in eliciting ambiguity attitude due to the concern of ambiguity hedge (Raiffa (1961)). Baillon, Halevy, and Li (2022b) find experimental evidence of using the RIM to hedge ambiguity. Johnson, Baillon, Bleichrodt, Li, Van Dolder, and Wakker (2021), and Baillon, Halevy and Li (2022a, 2022b) further suggest that randomization needs to precede the resolution of uncertainty so that incentive compatibility can be partially restored. In the current study, we adopt RIM as it facilitates within-subject comparison. Moreover, in both experiments, the subjects are only allowed to choose one color to bet on for all the lotteries in one choice list, making ambiguity hedge more difficult.

0; (4) longshot hazard: 10% chance of losing RMB50, otherwise 0; (5) mixed lottery: 50% chance of winning RMB100, otherwise losing RMB50. For each of the three types of uncertainty, the corresponding five lotteries are included, which results in a total number of 20 lotteries. The detailed implementation of each lottery is as follows.

For the three risk lotteries involving 50% chance, that is, moderate prospect, moderate hazard, and mixed lottery, subjects bet on the parity (odd versus even) of the number on a card randomly drawn from a deck of 10 cards of 5 odd and 5 even numbers. The corresponding three ambiguity lotteries concern betting on a deck of 10 cards with unknown composition of odd and even numbers. The three respective compound lotteries involve betting on a deck of 10 cards constructed with each of the 10 cards independently drawn from another deck of 10 cards of 5 odd and 5 even numbers. For the two risk lotteries involving 10% chance, that is, longshot prospect and longshot hazard, subjects bet on the exact number of a card randomly drawn from a deck of 10 cards numbered from 1 to 10. The corresponding two ambiguity lotteries are implemented by betting on a deck of 10 cards with unknown composition of numbers from 1 to 10. The two respective compound lotteries are implemented by betting on a deck of 10 cards constructed with each of the 10 cards independently drawn from another deck of 10 cards numbered from 1 to 10. At last, the five natural-event lotteries are based on market closing of the Tehran Stock Exchange Index on a given day. Subjects bet on the parity or the exact number of the second decimal digit of the close price for the lotteries with moderate or longshot likelihoods, respectively. Similar to Experiment 1, to ameliorate potential effects of suspicion, we allow subjects to choose freely the parity or exact number to bet on in all of the related lotteries.

To elicit the CE of each lottery, we again adopt the price-list design with subjects choosing between a given lottery and a sure amount from a list of 21 amounts. The list of sure amounts are arranged in an ascending order with the expected payoff of the lottery in the middle. Some of the sure amounts are chosen to first order stochastically dominate or be dominated by the lottery. More specifically, for moderate prospect (50% chance of winning RMB100, otherwise 0), we include 0 and RMB100 as two of the sure amounts. For moderate hazard (50% chance of losing RMB50, otherwise 0), we include RMB50 and 0 as two of the sure amounts. For both longshot prospect (10% chance of winning RMB100, otherwise 0), and longshot hazard (10% chance of losing RMB50, otherwise 0), we include RMB50 as one of the sure amounts (see Appendix B for details).

In order to avoid a substantial increase in the overall cognitive effort demanded from our subjects, we do not include duplicate elicitations for each of 20 lotteries as in Experiment 1. As a result, we cannot implement the ORIV for the data analysis in Experiment 2. In each price list, subjects make 21 binary choices, resulting in a total of 420 choices for 20 lotteries. To further simplify the choice tasks, we enforce a single switching point for each choice list and subjects only need to click on one sure amount in each list indicating that they would choose the lottery over the sure amounts that are strictly lower than their chosen amount and switch to the lottery for the rest. The sure amount chosen serves as the switch point and provides a proxy for CE of the lottery.

We randomize the order of the 20 lists in a two-step manner. First, within each of the five types of lotteries varying in outcomes and likelihoods, the risk lotteries always appear first, and the three types of uncertainty appear later in random order. Second, we randomize the order of the five lotteries. As with Experiment 1, we adopt the random incentive mechanism, paying each subject based on one of her randomly selected decisions in the experiment. To incentivize participation and to avoid subjects losing money out of their own pocket, each subject receives a show-up fee of RMB60. If the randomly selected decision leads to a loss of RMB50, 50 will be deducted from the show-up fee. The experiment is implemented in o-tree, an open-source platform for behavioral research. We have recruited 208 undergraduate students from Wuhan University as participants using e-advertisement (see Appendix B for experimental instructions).

4. Results

In this section, we first present the results from Experiment 1 on the links among attitudes toward the three types of uncertainty based on even-chance gain bets, including summary statistics, the effects of cognitive ability, and replication in Beijing. We end the section with the generalizability of the results of Experiment 1 to include small probability and loss outcome in Experiment 2.

The summary statistics of Experiment 1 is presented in Table 4 for the eight lotteries (see also Figure A.1 and Figure A.2 in Appendix A in the Online Supplementary Material (Chew, Miao, and Zhong (2023)). The data is coded as the number of choices in which subjects choose the lottery over the sure amount. A higher number means a higher CE with number 7 indicating risk neutrality. We summarize the number of observations, the mean and standard deviation of CE, and the percentage of subjects who are neutral, averse, and seeking for the baseline risk lottery.¹⁴ We also compare the CE of the risk lottery to that of the rest of the lotteries, and report the proportion of subjects who are neutral, averse and seeking toward ambiguity, compound risk, and natural-event uncertainty.

We observe that the majority of the subjects are ambiguity averse. Among the ambiguity lotteries, the CE of the full ambiguity lottery: CE_A is significantly lower than that of the interval ambiguity lottery: CE_{A1} (paired t-test, p < 0.001), as well as that of the

	Lottery	Ν	Mean	SD	Neutral	Aversion	Seeking
R	Risk	1956	5.239	2.595	9.4%	76.0%	14.7%
Α	Full ambiguity	2031	2.910	3.041	22.0%	65.0%	13.0%
Al	Interval ambiguity	2014	3.683	3.003	25.2%	55.8%	19.1%
A2	Disjoint ambiguity	2011	3.989	3.364	23.2%	53.6%	23.2%
С	Full compound risk	2016	3.672	3.246	23.1%	56.3%	20.6%
Cl	<i>p-q</i> compound risk	2018	4.679	3.383	23.8%	44.3%	32.0%
Ν	Natural event (home)	2025	4.962	2.771	35.9%	35.4%	28.7%
N1	Natural event (foreign)	2022	4.659	2.788	35.3%	39.8%	24.9%

TABLE 4. Summary statistics of attitudes toward uncertainty.

¹⁴We observe 2.4% to 6% of choice lists with multiple switch points and exclude these data from the analysis. This results in the difference in number of observations.

disjoint ambiguity: CE_{A2} (paired t-test, p < 0.001). These findings replicate the corresponding results in Chew, Miao, and Zhong (2017).

For the compound lotteries, consistent with the findings in previous experimental studies (e.g., Abdellaoui, Klibanoff, and Placido (2015)), we observe that the majority of the subjects exhibit compound-risk aversion. Between the two compound lotteries, the CE of the full compound lottery: CE_C, is significantly lower than that of the *p*-*q* compound lottery: CE_{C1} (paired t-test, *p* < 0.001). This may reflect the sense that *p*-*q* compound lottery is less complex in the sense of having less states in the stage-1 risk compared to the full compound lottery.

For natural-event lotteries, while about one-third of the subjects exhibit neutrality toward the two temperature lotteries, more subjects are averse to natural-event uncertainty, especially for the foreign temperature lottery. Between these two lotteries, the CE of home temperature lottery CE_N is significantly higher than that of the foreign temperature lottery: CE_{N1} (paired t-test, p < 0.001). Using natural-event lotteries with almost objective uncertainty, we demonstrate a preference for familiarity in a within-subject design. This extends the findings of previous studies on preference for familiarity whereby source preference can also be attributable to potential difference in terms of knowledge (Fox and Tversky (1995), Abdellaoui et al. (2011)).¹⁵

4.1 New links: Ellsberg meets Keynes

We examine the links among attitudes toward ambiguity, compound risk, and almostobjective natural events. Table 5. A reveals a significant association between ambiguity nonneutrality and non-RCLA behavior (Pearson's chi-squared test, p < 0.001). Of 203 subjects who are ambiguity neutral, 124 of them exhibit compound risk neutrality. This is more than six times the expected frequency under the null hypothesis of independence and replicates the studies on the link between ambiguity and compound lottery since Halevy (2007).

Table 5.B reveals a significant association between ambiguity neutrality and neutrality toward natural events (Pearson's chi-squared test, p < 0.001). Specifically, of 202 ambiguity neutral subjects, 153 subjects are neutral toward natural events, which is more than 2.5 times the expected frequency under the null hypothesis of independence. This unveils a missing link between ambiguity attitude and attitude toward natural-event uncertainty. Moreover, Table 5.C shows a significant association between RCLA behavior and neutrality toward natural-event uncertainty (Pearson's chi-squared test, p < 0.001). Specifically, of 192 subjects exhibiting RCLA, 148 subjects are neutral toward naturalevent uncertainty, which is about 3 times the expected frequency under the null hypothesis of independence. This reveals another missing link between attitude toward compound risk and attitude toward natural-event uncertainty.

¹⁵We also examine the effect of background characteristics including gender, age, parental education, number of siblings, and family income (Table A.1 for Singapore subjects and A.2 for Beijing subjects). We find that female subjects are more risk averse, compared to the male subjects, and subjects from lower income families are more risk averse, relative to those from higher income families. We do not observe significant and consistent effects of other characteristics on attitudes toward risk, ambiguity, compound risk, and natural-event uncertainty.

	A. Ambiguity and co	mpound risk		
	Compound risk			
Ambiguity	Reduction	Nonreduction	Total	
Neutral	124 (20.7)	79 (182.3)	203	
Nonneutral	67 (170.3)	1607 (1503.7)	1674	
Total	191	1686	1877	
Pearson chi2(1) = 645.38 ,	p < 0.001			
	B. Ambiguity and na	atural events		
	Natur	al events		
Ambiguity	Neutral	Nonneutral	Total	
Neutral	153 (62.2)	49 (139.8)	202	
Nonneutral	426 (516.80)	1253 (1162.2)	1679	
Total	579	1302	1881	
Pearson $chi2(1) = 214.71$,	p < 0.001			
	C. Natural events and	compound risk		
	Natur	al events		
Compound risk	Reduction	Nonreduction	Total	
Neutral	148 (58.7)	44 (133.3)	192	
Nonneutral	430 (519.3)	1267 (1177.7)	1697	
Total	578	1311	1889	
Pearson chi2(1) = 217.48 ,	p < 0.001			

TABLE 5.	Linking attitudes	toward three	types of	uncertainty.
INDEL O.	Linning attitudeo	tomara unce	<i>cypco</i> or	anoon tunney.

Note: This two-way table presents the number of subjects by (non)neutrality toward ambiguity, compound risk, and natural-event uncertainty. Each cell indicates the number of subjects with the expected number displayed in parentheses (Pearson's chi-squared test, p < 0.001).

We further investigate the association among the three uncertainty attitudes by assessing the correlations among the ambiguity premium, the compound risk premium and the natural-event uncertainty premium. The respective premiums are measured by the difference in CEs between the risk lottery and the ambiguity (compound/naturalevent) lottery. We find that the correlations between premiums across each individual lottery mostly lie between 0.40 and 0.60 (Table A.3 in Appendix A). The pairwise scatter plots of the average premium in each of the three types of uncertainty are presented in Figure 1. We observe significant correlations between average ambiguity premium and average compound risk premium (0.62), between average ambiguity premium and average natural-event uncertainty premium (0.49), and between average compound risk premium and average natural-event uncertainty premium (0.48).

A common concern in preference elicitation is measurement error. Note that the observed correlations are highly significant among the three ambiguity lotteries (0.75, 0.56, and 0.63), between the two compound lotteries (0.49), and between the two naturalevent lotteries (0.87), This suggests common underpinnings within each type of uncertainty in conjunction with measurement errors. In subsequent analysis, we make use of the method of Obviously Related Instrumental Variables (ORIV) developed in Gillen,



FIGURE 1. Pairwise scatter plots of average premium across three types of uncertainty.

Snowberg, and Yariv (2019) to correct for measurement error given that we have multiple elicitations of closely related attitudes.¹⁶ Specifically, we have three ambiguity lotteries, two compound lotteries, and two natural-event lotteries that can serve as multiple elicitations. As reported in Table 1 in the Introduction, the correlations are substantially improved after adopting ORIV: correlation between ambiguity premium and compound risk premium increases to 0.83, correlation between ambiguity premium and natural-event uncertainty premium increases to 0.58, and correlation between compound risk premium and natural-event uncertainty premium increases to 0.66.¹⁷ This is in line with the observations in Gillen, Snowberg, and Yariv (2019), which reports an increase in the correlation between ambiguity premium from 0.44 to 0.85 using ORIV. Overall, these analyses further strengthen the links among these three uncertainty attitudes.

Finally, we examine the behavior at the individual level (Table 6). A subject is classified as being neutral toward a particular type of uncertainty if the CEs for lotteries in that type of uncertainty are the same as the CE for the risk lottery. We find that 65.3% of the subjects are nonneutral toward all three types of uncertainty, while 5.7% of the subjects are neutral toward all three. For the rest of subjects, 20.4% are nonneutral toward ambiguity and compound risk, and neutral toward natural-event uncertainty. These three

¹⁶ORIV adopts an errors-in-variable instrumental variable approach to analyze the data when there are multiple elicitations of the behavioral proxies.

¹⁷In addition, we also use principal component analysis within each source of uncertainty and use the first component to conduct the correlation analysis. The correlation is 0.64 between ambiguity premium and compound risk premium, 0.54 between ambiguity premium and natural-event uncertainty premium, and 0.51 between compound risk premium and natural-event uncertainty premium.

Ambiguity	Compound risk	Natural event	Percent
Nonneutral	Nonneutral	Nonneutral	65.3%
Nonneutral	Nonneutral	Neutral	20.4%
Neutral	Neutral	Neutral	5.7%
Nonneutral	Neutral	Nonneutral	1.4%
Neutral	Nonneutral	Nonneutral	1.7%
Nonneutral	Neutral	Neutral	2.2%
Neutral	Nonneutral	Neutral	2.5%
Neutral	Neutral	Nonneutral	0.9%

TABLE 6. 1	Individual (type analysis.
------------	--------------	----------------

Note: This table presents the percentage of individual type classified by attitudes toward three types of uncertainty. Neutrality toward certain type of uncertainty corresponds to the same CE between the risky lottery and each of the lotteries in that type of uncertainty.

groups account for 91% of the subjects, with the rest of 9% exhibiting other behavioral patterns.

4.2 Cognitive ability

In this subsection, we explore the role of cognitive ability measured by subjects' scores in Raven's test (mean = 56.02, median = 57; SD = 4.60) by comparing subjects with high scores and those with low scores (1059 subjects with scores higher than 57; 1022 subjects with scores lower than or equal to 57). We report the statistics in Tables A.4 to A.6 in Appendix A, and summarize the results here. Overall, our results are robust with respect to cognitive ability. For both high ability and low ability groups, we find: (1) correlation remains high across all three types of uncertainty and is higher between attitudes toward ambiguity and compound risk; (2) a strong majority of the subjects are nonneutral toward the three types of uncertainty, followed by subjects who are nonneutral toward ambiguity and compound risk, but neutral toward natural-event uncertainty, with neutrality toward all three types coming in at a distant third.

We further examine the role of cognitive ability. First examining how cognitive ability is related to attitude toward ambiguity, compound risk, and natural-event uncertainty, we find one out of seven lotteries exhibiting significant difference. More specifically, subjects with high scores tend to have a higher premium for full ambiguity, compared with those with low scores (OLS, p = 0.033). We then examine the pairwise correlations across the three types of uncertainty, and find that above-median subjects are linked to higher correlation between ambiguity attitude and compound risk attitude, and lower correlation between ambiguity attitude and natural-event uncertainty attitude, as well as between compound risk attitude and natural-event uncertainty attitude. Lastly, in individual type analysis, among subjects exhibiting nonneutrality toward all three types of uncertainty, the percentage of 63.4% for above-median subjects is significantly lower than the 67.4% for below-median subjects (proportion test, p < 0.067). For subjects exhibiting neutrality toward all three types of uncertainty, the percentage of uncertainty, the percentage is 6.6% for above-median subjects and 4.6% for below-median subjects (proportion test, p < 0.070). In

sum, close links are observed across types of uncertainty with respect to cognitive ability. Moreover, subjects with high scores are less likely to be nonneutral toward all three types of uncertainty. We would also like to add a cautious note that none of the correlations between IQ and attitudes toward the seven lotteries remains statistically significant after we correct for the multiple comparisons conducted when analyzing the seven lotteries.

4.3 Replicability

In this subsection, we examine the replicability of our findings using the Beijing sample. We display the results of our analysis in Tables A.7 to A.9 in Appendix A and summarize them here. Overall, the high correlations among attitudes toward the three types of uncertainty, and the similarity in proportions of individual types, are replicated in the Beijing sample. To explore potential differences between the Beijing and Singapore samples, we first examine subjects' attitudes toward ambiguity, compound risk, and naturalevent uncertainty. Despite statistical significance, the differences in proportion between Singapore and Beijing samples of being averse are generally less than 3%. One exception is with the p-q compound lottery, in which the proportion of being averse is 44.3% among Singapore subjects compared to 31.5% among Beijing subjects. We then examine the pairwise correlations across the three types of uncertainty, and find that the correlation is 0.78 between ambiguity attitude and compound risk attitude, 0.60 between ambiguity attitude and natural-event uncertainty attitude, and 0.62 between compound risk attitude and natural-event uncertainty attitude. Furthermore, in individual type analysis, 68.5% of Beijing subjects exhibit nonneutrality toward all three types of uncertainty, 16.5% exhibit nonneutrality toward ambiguity and compound risk, and neutrality toward natural-event uncertainty, and 4.0% are neutral toward all three types of uncertainty. Lastly, the observed patterns with regard to cognitive ability are also replicated in Beijing sample (see Tables A.10 to A.12 in Appendix A). Overall, the choice behavior of Beijing subjects and Singapore subjects are remarkably similar. This said, Beijing subjects are less averse to p-q compound lottery, exhibit lower correlation between ambiguity attitude and compound risk attitude, and less likely to exhibit nonneutrality toward ambiguity and compound risk together with neutrality toward natural-event uncertainty.

4.4 Extendibility to fourfold and mixed lotteries

This subsection examines whether the links among the attitudes toward the three types of uncertainty observed in Experiment 1 can be generalized via Experiment 2 to lotteries in the fourfold pattern and additionally a mixed lottery involving both a gain and a loss outcome. The summary statistics for the 20 lotteries in Experiment 2 is presented in Table A.13.¹⁸ When comparing the CEs of risk lotteries with their expected values,

¹⁸We report the results after excluding 59 subjects who violate first-order stochastic dominance at least once. In the Appendix, we report the corresponding tables with the full sample of 208 subjects (Table A.14 and A.15). The results are quantitatively similar.

	A–C	A–N	C–N
Moderate prospect	0.556	0.398	0.504
Moderate hazard	0.669	0.492	0.548
Longshot prospect	0.687	0.602	0.637
Longshot hazard	0.613	0.584	0.627
Mixed lottery	0.588	0.617	0.615
Average	0.623	0.538	0.586

TABLE 7. Correlations across three types of uncertainty in Experiment 2.

Note: This table reports the Spearman correlations after dropping those subjects violating FOSD. N = 149.

we find that the majority of the subjects are risk averse for moderate prospect, longshot hazard, and mixed lottery, and are risk-seeking for moderate hazard and longshot prospect. In this regard, we replicate the observed fourfold pattern in risk attitude and aversion to mixed risk (Kahneman and Tversky (1979), Tversky and Kahneman (1992)). When comparing the CEs of risk lotteries with those of the other three types of uncertainty, we observe that subjects are, on average, averse to ambiguity, compound risk, and natural-event uncertainty except for longshot hazard. While it has been hypothesized that ambiguity attitudes exhibit a fourfold pattern, namely ambiguity aversion for moderate prospect and longshot hazard, and ambiguity seeking for moderate hazard and longshot prospect, experimental evidence has been mixed (see Trautmann and Van De Kuilen (2015) for a comprehensive review).¹⁹

Table 7 presents the correlations among the ambiguity premium, the compound risk premium and the natural-event uncertainty premium. Similarly, the respective premiums are measured by the difference in CEs between the risk lottery and the ambiguity (compound/natural-event) lottery. Notice that we do not include multiple elicitations of closely related lotteries in Experiment 2, so we cannot apply ORIV to correct measurement errors. Nevertheless, we find that the correlations of premiums across each individual lottery and different types of uncertainty range between 0.40 and 0.69 with an average of 0.58. In particular, for all the five lotteries considered, the average correlation is 0.62 between ambiguity premium and compound risk premium, 0.54 between ambiguity premium and natural-event uncertainty premium. Overall, these results provide strong support for the existing link in Halevy (2007) and the new links discovered in Experiment 1.

5. Discussion

Related experimental studies

Our results have bearing on recent experimental studies testing ambiguity models both qualitatively and quantitatively. Some experimental studies focus exclusively on mod-

¹⁹For example, while Baillon and Bleichrodt (2015) provide experimental support for the fourfold pattern, they also observe ambiguity neutrality for unlikely events. Relatedly, Kocher, Lahno, and Trautmann (2018) observe ambiguity neutrality or ambiguity seeking for moderate hazard and longshot prospect, and ambiguity seeking for moderate hazard.

els involving multiple priors. For example, Conte and Hey (2013) examine the predictive power of multiple-prior models through estimating a mixture model, and find that recursive expected utility explains the overall patterns better than α -maxmin expected utility. Cubitt, Van de Kuilen, and Mukerji (2020) test preference for hedging under ambiguity to discriminate between α -maxmin expected utility and recursive expected utility, and find support for the latter. When different outcomes and different levels of likelihoods are considered, Baillon and Bleichrodt (2015) find different ambiguity attitudes toward gains and losses, and toward moderate and longshot likelihoods. They suggest that their findings are more consistent with prospect theory (adapted to the domain of ambiguity), but to a lesser extent, with α -maxmin expected utility or recursive expected utility.

Building on the works of Tversky and his colleagues (Heath and Tversky (1991), Fox and Tversky (1995)), a number of recent studies examine attitude toward natural-event uncertainty. de Lara Resende and Wu (2010) show that subjects' preference for betting on a natural event as well as its complement, such as the temperature of a particular city being higher or lower than a given level, may depend on their level of knowledge of the specific source of uncertainty. Adopting the definition of event exchangeability in Chew and Sagi (2008), Abdellaoui et al. (2011) use a bisection procedure for subjects to partition the state space, for example, the range of home (foreign) city temperature, into disjoint intervals with equal likelihoods. They observe that a substantial proportion of subjects exhibit a preference for betting on the temperature in their home city rather than a foreign city. Relating to the opening quote of Keynes, subjects in their study may possess different degrees of confidence or affinity with the source-home versus foreign-of the underlying temperature uncertainty, despite having assessed their likelihoods as being equal. In this design, models with multiple priors can still account for the observed preference for the familiar with different (distributions on the) sets of priors based on the degree of confidence in the assessed probability. By contrast, the current study is the first to investigate source preference using almost-objective uncertainty and to examine its interaction with compound risk attitude and ambiguity attitude.

Almost objective uncertainty

Machina (2004) proposes the notion of "almost objective" uncertainty, and argues that most randomization devices, including a simple toss of a coin, may actually generate almost-objective uncertainty. In Experiment 1, the uncertainty associated with the trailing digit of a city temperature being odd or even is almost-objective. Yet, we observe a preference for the risk lottery to these two almost-objective lotteries, as well as a significant but weaker preference to bet on home temperature rather than on foreign temperature. In Experiment 2, when we use the second decimal digit at market closing of a stock exchange index, we observe similar preference for the risk lottery to the almost-objective lottery. These observations challenge ambiguity models adopting the multiple-prior perspective, whose preferences over almost-objective acts approach probabilistic sophistication asymptotically. In contrast, the source preference perspective is compatible with the wide range of phenomena mentioned here through decision makers having preference over distinct sources of uncertainty.

In the meantime, we cannot rule out the possibility of some subjects having different degrees of confidence in assessing the likelihood in terms of home versus foreign city temperature. A Singaporean subject may have heard nothing (prior) bizarre about local temperature and updates her belief and treats the (posterior) probability of the trailing digit being odd/even as almost objectively 50-50. She may not be as confident when updating her posterior about the foreign city temperature given a weaker prior. This reasoning may rely on an implicit assumption that the natural event is "perceived" as embracing a certain degree of ambiguity despite its almost-objective nature. In the example in Section II, while the subjective probability of the union of left-half of each interval $\bigcup_n (\frac{n}{m}, \frac{n+0.5}{m}]$ approaches 0.5 asymptotically, a subject may think of such a way of partitioning the state space cognitively demanding and simply partition the state space in to {odd, even}, thereby generating a certain degree of ambiguity. An interesting question concerns whether this as-if view should be modeled through the lens of multiple priors or source preference. While either way is admissible, we note that our theoretical predictions in Section II need to be revised should we adopt this *as-if* view in conjunction with the multiple-prior perspective. In particular, all of the models considered there, especially those adopting the multiple-prior perspective, would be compatible with natural-event nonneutrality, and can then generate a correlation between almostobjective uncertainty attitude and ambiguity attitude, or even compound risk attitude, depending on whether a two-stage approach is adopted.

Rich domains of uncertainty

A number of recent studies can be interpreted as investigating the commonalities of attitudes across types of uncertainty. Armantier and Treich (2015) consider a "complex risk" treatment, in which the outcome of the bet depends on the color of two balls simultaneously drawn from two risky urns. They observe a tight association between attitude toward this complex risk and the attitude toward ambiguity, and suggest that their finding may hint at the link between ambiguity attitude and complexity attitude. Epstein and Halevy (2018) examine bets, which depend on the color of two balls drawn from two urns, where there is ambiguity in the compositions of each urn as well as the correlation between them. They find that attitude toward such ambiguous correlation between two unknown urns is related to attitude toward the standard Ellsbergian ambiguity. Li, Muller, Wakker, and Wang (2018) examine uncertainty arising from betting on the home district of an Indian rural child. Given that subjects have no prior knowledge on different Indian districts, they observe a significant correlation between attitudes toward betting on the districts and betting on an unknown urn.

In their seminal work of prospect theory, Kahneman and Tversky (1979) posit that attitudes toward risk are sensitive to outcomes and likelihoods, and arrive at the so-called fourfold pattern in risk attitude: risk aversion toward moderate prospect or longshot hazard, and risk seeking toward longshot prospect or moderate hazard. The fourfold pattern in risk attitude has been widely observed in experimental studies on decisionmaking under risk. Correspondingly, in the literature on decision-making under uncertainty, a number of experimental studies investigate ambiguity attitude beyond the even-chance prospect and find mixed evidence regarding how ambiguity attitude may depend on outcome and likelihood (i.e., de Lara Resende and Wu (2010), Baillon and Bleichrodt (2015), Kocher, Lahno, and Trautmann (2018)). In Experiment 2, we observe a reassuring fourfold pattern in risk attitude as well as an aversion to even-chance mixed risk. In addition, we observe (weak) aversion toward all three types of uncertainty except for longshot hazard. While attitudes toward risk and ambiguity depend on outcome and likelihood, our study shows that the pairwise correlations among attitudes toward the three types of uncertainty do not exhibit such sensitivity. The robustness in the observed links suggests a shared component in decision-making underpinning different attitudes toward the three different types of uncertainty.

Cognitive and consequentialist considerations

Implicit in the formulation of the expected utility hypothesis is an exclusive focus on what may be considered as consequences in an uncertain alternative, chiefly its outcomes and their associated likelihoods. This consequentialist focus is also evident in the development of nonexpected utility models, culminating in the definition of probabilistic sophistication in Machina and Schmeidler (1992). It has been suggested that departures from the prescriptions of expected utility, such as Allais and Ellsbergian behavior, exemplify nonconsequentialist behavior and may have their roots in limitations in cognitive ability or attentiveness (see, e.g., Morgenstern (1979)). To address this cognitive limitation hypothesis in conjunction with ambiguity attitude elicited using the two-urn problem, subjects in Chew, Ratchford, and Sagi (2017) are first screened for comprehension and attentiveness before being presented with a matrix version of the two-urn problem intended to minimize the difference in complexity between the known and the unknown bets. They find that low-comprehension subjects are seemingly ambiguity neutral in choosing randomly while high-comprehension subjects continue to favor the known bet.²⁰

As discussed earlier, 57 out of 60 is the median score of our subjects in Ravens IQ test following the choice tasks. This suggests that the bulk of our subjects are high-comprehension and attentive. At the same time, we continue to observe systematic departures from consequentialist behavior in attitudes toward the three types of uncertainty, including the strong pairwise correlations, among the subsample of above-median subjects.²¹ In conjunction with the literature, our findings reveal that decision-making under uncertainty can depend on how choice is perceived even when there is sufficient attention and comprehension.

²⁰It is instructive to revisit Raiffa's (1961) reflection on his "error" in favoring the known bet when responding to Ellsberg's request to provide his choice in the two-urn problem "without any pencil pushing." He realized afterwards that he could fashion from the unknown bet a two-stage lottery with overall winning probability of 1/2 by betting red if a coin flip turns out head and black otherwise. Consistent with RCLA, Raiffa would be indifferent between the two-stage bet and the known bet. Raiffa's reflection points to an innate tendency to avoid ambiguity when he is not particularly attentive while there is little doubt about his ability to behave in a consequentialist manner with sufficient attention.

²¹In this regard, the finding in Abdellaoui, Klibanoff, and Placido (2015) of the ambiguity-compound risk link being stronger for nonengineers than engineers may be due to greater engagement and attentiveness.

Toward an integrative theory

Despite the flexibility in viewing each type of uncertainty as a distinct source, a closer examination of existing models adopting the source preference perspective reveals that they do not deliver a joint account of our experimental findings. In particular, the Chew-Sagi approach admits RCLA and cannot distinguish compound risk from its reduced simple risk, thereby failing to generate the observed links between compound risk attitude and ambiguity attitude or natural-event attitude. On the other hand, Nau (2006) and SPS limit their domain to two exogenous sources of uncertainty and coincide with recursive models upon interpreting the two sources as priors at different stages. As such, they fail to generate nonneutral attitude toward almost-objective natural event as well as the observed links between natural-event attitude and ambiguity attitude or compound risk attitude. Overall, the observed differences in attitudes toward three types of uncertainty as well as the pairwise links among these uncertainty attitudes point to the possibility of having an integrative theory across the multiple-prior and the source perspectives. One may consider for simplicity a Chew-Sagi type of source model without RCLA that treats compound risk as yet another source of uncertainty, which is different from its reduced simple risk. However, such a model lacks tractability, and more importantly, cannot offer a plausible account for the stronger correlation in attitudes between ambiguity and compound risk than that between ambiguity and natural-event uncertainty as well as between compound risk and natural-event uncertainty. This stronger correlation between ambiguity attitude and compound risk attitude tends to support the recursive approach to modeling ambiguity attitude.²² Taken together, our findings point to an integrative theory, which extends the current recursive models by incorporating within-stage source preferences.

6. CONCLUSION

The consideration of the unknown urn by Keynes and Ellsberg has spawned a voluminous literature on decision-making under uncertainty encompassing theory, experiment, and application, and spanning multiple disciplines. Subsequently, distinct perspectives concerning betting on the unknown urn has given rise to two parallel strands of research. In one strand of research, ambiguity attitude reflects Ellsberg's multipleprior perspective and can be linked to attitude toward objective compound risk. In the other strand, ambiguity attitude arises from having preference over the sources of uncertainty underpinning a lottery, which can range from urns and balls to city temperatures and market indices.

²²The observed link between attitudes toward full ambiguity and full compound risk in Experiment 1 admits an alternative interpretation that the compound lottery may be viewed as ambiguity due to the similarity in presentation or its complexity. We observe an equally significant correlation in attitudes between full ambiguity and p-q compound risk as that between full ambiguity and full compound risk. As it is unlikely for the subjects to view the p-q compound lottery as ambiguity, our results favor the view that subjects view ambiguity as compound lottery. In this regard, our findings lend support to the recursive modelling approach, and corroborate Halevy's (2007) observed close link between ambiguity attitude and compound risk attitude.

The present paper examines experimentally attitudes toward multiple-prior uncertainty in terms of ambiguity and its corresponding objective compound risk and attitude toward almost-objective uncertainty. We observe a general tendency to be nonneutral toward each of the three types of uncertainty and each that pair of these attitudes tend to be highly correlated. We further find that this tendency is robust with respect to the valence of the outcomes and to the degree of likelihood. In this regard, attitude toward one type of uncertainty may be informative in predicting attitude toward another type of uncertainty.

Our findings contribute to an emerging sense of the role of perception in decisionmaking under uncertainty. Whether a bet on the unknown urn is perceived as being probabilistic (as with Keynes) or having multiple priors (as with Ellsberg), can materially impact how it is evaluated. Taken as a whole, our findings point to the need for an integrative model of decision-making under uncertainty to incorporate how decision makers may perceive a lottery not only in terms of its likelihoods and the valence of its outcomes, but also its underlying sources of uncertainty as well as whether and how attitudes toward such uncertainties may be linked.

References

Abdellaoui, Mohammed, Aurélien Baillon, Laetitia Placido, and Peter P. Wakker (2011), "The rich domain of uncertainty: Source functions and their experimental implementation." *American Economic Review*, 101, 695–723. [1141, 1148, 1154]

Abdellaoui, Mohammed, Peter Klibanoff, and Laetitia Placido (2015), "Experiments on compound risk in relation to simple risk and to ambiguity." *Management Science*, 61, 1306–1322. [1135, 1142, 1148, 1156]

Alon, Shiri and David Schmeidler (2014), "Purely subjective maxmin expected utility." *Journal of Economic Theory*, 152, 382–412. [1139]

Anscombe, Francis J. and Robert J. Aumann (1963), "A definition of subjective probability." *Annals of mathematical statistics*, 34, 199–205. [1139]

Armantier, Olivier and Nicolas Treich (2015), "The rich domain of risk." *Management Science*, 62, 1954–1969. [1155]

Azrieli, Yaron, Christopher P. Chambers, and Paul J. Healy (2018), "Incentives in experiments: A theoretical analysis." *Journal of Political Economy*, 126, 1472–1503. [1145]

Bade, Sophie (2015), "Randomization devices and the elicitation of ambiguity-averse preferences." *Journal of Economic Theory*, 159, 221–235. [1145]

Baillon, Aurélien and Han Bleichrodt (2015), "Testing ambiguity models through the measurement of probabilities for gains and losses." *American Economic Journal: Microeconomics*, 7, 77–100. [1153, 1154, 1156]

Baillon, Aurélien, Han Bleichrodt, Chen Li, and Peter P. Wakker (2021), "Belief hedges: Measuring ambiguity for all events and all models." *Journal of Economic Theory*, 198, 105353. [1136] Baillon, Aurélien, Yoram Halevy, and Chen Li (2022a), "Experimental elicitation of ambiguity attitude using the random incentive system." *Experimental Economics*, 1–22. [1145]

Baillon, Aurélien, Yoram Halevy, and Chen Li (2022b), "Randomize at your own risk: On the observability of ambiguity aversion." *Econometrica*, 90, 1085–1107. [1145]

Barseghyan, Levon, Jeffrey Prince, and Joshua C. Teitelbaum (2011), "Are risk preferences stable across contexts? Evidence from insurance data." *American Economic Review*, 101, 591–631. [1137]

Becker, Selwyn W. and Fred O. Brownson (1964), "What price ambiguity? Or the role of ambiguity in decision-making." *Journal of Political Economy*, 72, 62–73. [1134]

Berger, Alex and Agnieszka Tymula (2022), "Controlling ambiguity: The illusion of control in choice under risk and ambiguity." *Journal of Risk and Uncertainty*, 65, 261–284. [1144]

Burks, Stephen V., Jeffrey P. Carpenter, Lorenz Goette, and Aldo Rustichini (2009), "Cognitive skills affect economic preferences, strategic behavior, and job attachment." *Proceedings of the National Academy of Sciences*, 106, 7745–7750. [1145]

Casadesus-Masanell, Ramon, Peter Klibanoff, and Emre Ozdenoren (2000), "Maxmin expected utility over Savage acts with a set of priors." *Journal of Economic Theory*, 92, 35–65. [1139]

Charness, Gary, Aldo Rustichini, and Jeroen Van de Ven (2011), "Self-confidence and strategic deterrence." Working Paper. [1145]

Chew, Soo Hong, Bin Miao, and Songfa Zhong (2017), "Partial ambiguity." *Econometrica*, 85, 1239–1260. [1135, 1142, 1148]

Chew, Soo Hong, Mark Ratchford, and Jacob S. Sagi (2017), "You need to recognize ambiguity to avoid it." *The Economic Journal*, 128, 2480–2506. [1156]

Chew, Soo Hong and Jacob S. Sagi (2008), "Small worlds: Modeling preference over sources of uncertainty." *Journal of Economic Theory*, 139, 1–24. [1135, 1140, 1154]

Chew, Soo Hong, Bin Miao, and Songfa Zhong (2023), "Supplement to 'Ellsberg meets Keynes at an urn'." *Quantitative Economics Supplemental Material*, 14, https://doi.org/10.3982/QE2253. [1138, 1147]

Conte, Anna and John D. Hey (2013), "Assessing multiple prior models of behaviour under ambiguity." *Journal of Risk and Uncertainty*, 46, 113–132. [1154]

Cubitt, Robin, Gijs Van de Kuilen, and Sujoy Mukerji (2020), "Discriminating between models of ambiguity attitude: A qualitative test." *Journal of European Economic Association*, 18, 708–749. [1154]

de Lara Resende, José Guilherme and George Wu (2010), "Competence effects for choices involving gains and losses." *Journal of Risk and Uncertainty*, 40, 109–132. [1154, 1156]

Dean, Mark and Pietro Ortoleva (2019), "The empirical relationship between nonstandard economic behaviors." *Proceedings of the National Academy of Sciences*, 116, 16262– 16267. [1135]

Dillenberger, David and Uzi Segal (2017), "Skewed noise." *Journal of Economic Theory*, 169, 344–364. [1143]

Einav, Liran, Amy Finkelstein, Iuliana Pascu, and Mark R. Cullen (2012), "How general are risk preferences? Choices under uncertainty in different domains." *American Economic Review*, 102, 2606–2638. [1137]

Ellsberg, Daniel (1961), "Risk, ambiguity, and the Savage axioms." *Quarterly Journal of Economics*, 75, 643–669. [1133, 1134]

Epstein, Larry G. and Yoram Halevy (2018), "Ambiguous correlation." *Review of Economic Studies*, 86, 668–693. [1155]

Ergin, Haluk and Faruk Gul (2009), "A theory of subjective compound lotteries." *Journal of Economic Theory*, 144, 899–929. [1134, 1141]

Fox, Craig R. and Amos Tversky (1995), "Ambiguity aversion and comparative ignorance." *Quarterly Journal of Economics*, 110, 585–603. [1135, 1140, 1148, 1154]

Gajdos, Thibault, Takashi Hayashi, Jean-Marc Tallon, and Jean-Christophe Vergnaud (2008), "Attitude toward imprecise information." *Journal of Economic Theory*, 140, 23–56. [1134]

Ghirardato, Paolo, Fabio Maccheroni, and Massimo Marinacci (2004), "Differentiating ambiguity and ambiguity attitude." *Journal of Economic Theory*, 118, 133–173. [1134]

Gilboa, Itzhak (1987), "Expected utility with purely subjective nonadditive probabilities." *Journal of Mathematical Economics*, 16, 65–88. [1134, 1139]

Gilboa, Itzhak and David Schmeidler (1989), "Maxmin expected utility with non-unique prior." *Journal of Mathematical Economics*, 18, 141–153. [1134, 1139]

Gill, David and Victoria Prowse (2016), "Cognitive ability, character skills, and learning to play equilibrium: A level-k analysis." *Journal of Political Economy*, 124, 1619–1676. [1145]

Gillen, Ben, Eric Snowberg, and Leeat Yariv (2019), "Experimenting with measurement error: Techniques with applications to the caltech cohort study." *Journal of Political Economy*, 127, 1826–1863. [1135, 1136, 1144, 1149, 1150]

Halevy, Yoram (2007), "Ellsberg revisited: An experimental study." *Econometrica*, 75, 503–536. [1134, 1135, 1140, 1142, 1148, 1153, 1157]

Halevy, Yoram and Vincent Feltkamp (2005), "A Bayesian approach to uncertainty aversion." *The Review of Economic Studies*, 72, 449–466. [1140]

Heath, Chip and Amos Tversky (1991), "Preference and belief: Ambiguity and competence in choice under uncertainty." *Journal of Risk and Uncertainty*, 4, 5–28. [1154] Johnson, Cathleen, Aurélien Baillon, Han Bleichrodt, Zhihua Li, Dennie Van Dolder, and Peter P. Wakker (2021), "Prince: An improved method for measuring incentivized preferences." *Journal of Risk and Uncertainty*, 62, 1–28. [1145]

Kahneman, Daniel and Amos Tversky (1979), "Prospect theory: An analysis of decision under risk." *Econometrica*, 47, 263–292. [1143, 1145, 1153, 1155]

Keynes, John Maynard (1921), A Treatise on Probability. MacMillan, London. [1133, 1134]

Klibanoff, Peter, Massimo Marinacci, and Sujoy Mukerji (2005), "A smooth model of decision making under ambiguity." *Econometrica*, 73, 1849–1892. [1134, 1140, 1141]

Kocher, Martin G., Amrei Marie Lahno, and Stefan T. Trautmann (2018), "Ambiguity aversion is not universal." *European Economic Review*, 101, 268–283. [1153, 1156]

Li, Zhihua, Julia Müller, Peter P. Wakker, and Tong V. Wang (2018), "The rich domain of uncertainty explored." *Management Science*, 64, 3227–3240. [1155]

Machina, Mark J. (2004), "Almost-objective uncertainty." *Economic Theory*, 24, 1–54. [1138, 1154]

Machina, Mark J. and David Schmeidler (1992), "A more robust definition of subjective probability." *Econometrica*, 60, 745–780. [1138, 1156]

Morgenstern, Oskar (1979), "Some reflections on utility." In *Expected Utility Hypotheses and the Allais Paradox*, 175–183, Springer, Dordrecht. [1156]

Nau, Robert F. (2006), "Uncertainty aversion with second-order utilities and probabilities." *Management Science*, 52, 136–145. [1141, 1157]

Pratt, John W. (1964), "Risk aversion in the small in the large." *Econometrica*, 32, 122–136. [1135]

Quiggin, John (1982), "A theory of anticipated utility." *Journal of Economic Behavior and Organization*, 3, 323–343. [1141]

Raiffa, Howard (1961), "Risk, ambiguity, and the Savage axioms: Comment." *Quarterly Journal of Economics*, 75, 690–694. [1145, 1156]

Schmeidler, David (1989), "Subjective probability and expected utility without additivity." *Econometrica*, 57, 571–587. [1134, 1139]

Segal, Uzi (1987), "The Ellsberg paradox and risk aversion: An anticipated utility approach." *International Economic Review*, 28, 175–202. [1134, 1139, 1140, 1143]

Segal, Uzi (1990), "Two-stage lotteries without the reduction axiom." *Econometrica*, 58, 349–377. [1140]

Seo, Kyoungwon (2009), "Ambiguity and second-order belief." *Econometrica*, 77, 1575–1605. [1134, 1139, 1140]

Smith, Vernon L. (1969), "Measuring nonmonetary utilities in uncertain choices: The Ellsberg urn." *Quarterly Journal of Economics*, 83, 324–329. [1134, 1135]

Trautmann, Stefan T. and Gijs Van De Kuilen (2015), "Ambiguity attitudes." *The Wiley Blackwell handbook of judgment and decision making*, 1, 89–116. [1153]

Tversky, Amos and Daniel Kahneman (1992), "Advances in prospect theory: Cumulative representation of uncertainty." *Journal of Risk and Uncertainty*, 5, 297–323. [1137, 1143, 1153]

Vieider, Ferdinand M., Mathieu Lefebvre, Ranoua Bouchouicha, Thorsten Chmura, Rustamdjan Hakimov, Michal Krawczyk, and Peter Martinsson (2015), "Common components of risk and uncertainty attitudes across contexts and domains: Evidence from 30 countries." *Journal of the European Economic Association*, 13, 421–452. [1137]

Wakker, Peter P. (1987), "Nonadditive probabilities and derived strengths of preferences." Unpublished Manuscript. [1139]

Wakker, Peter P. (2007), "Message to referees who want to embark on yet another discussion of the random-lottery incentive system for individual choice." https://personal. eur.nl/wakker/miscella/debates/randomlinc.htm. [1145]

Weber, Elke U., Ann-Renee Blais, and Nancy E. Betz (2002), "A domain-specific riskattitude scale: Measuring risk perceptions and risk behaviors." *Journal of Behavioral Decision Making*, 15, 263–290. [1137]

Co-editor Garance Genicot handled this manuscript.

Manuscript received 22 September, 2022; final version accepted 21 February, 2023; available online 23 February, 2023.